Author’s response to reviews

Title: A 26 year old women presenting with an intrapartum sacral stress fracture to our clinic: A case presentation

Authors:

Miguel Pishnamaz (mpishnamaz@ukaachen.de)
Richard Sellei (rsellei@ukaachen.de)
Roman Pfeifer (rpfeifer@ukaachen.de)
Philipp Lichte (plichte@ukaachen.de)
Hans C. Pape (hpape@ukaachen.de)
Philipp Kobbe (pkobbe@ukaachen.de)

Version: 4 Date: 6 October 2011

Author’s response to reviews: see over
Dear Editor,

enclosed I’m sending you the revised version of the manuscript: MS: 2024161085570519. All corrections are highlight with tracked changes.

Reviewer one mentioned: The description of the results of the MRI could be ambiguous from the text. We did correct the passage and underlined that there was just one MRI preformed and that there was no inflammatory or ischemic process seen.

Reviewer two mentioned grammatical corrections on page 1, 3 and 4.

These have been corrected like described:

1. Head line: Changed year against years
2. Page 3: Corrected ran one hour daily against performed running
3. Page 4: Changed anesthesia against anaesthesia

Besides that we also address the formatting requests from your Editorial Office.

Thank you for editorial use of our case report.

Sincerely yours,

Miguel Pishnamaz